Title: Cisplatin induced posterior reversible encephalopathy syndrome and successful retreatment in a patient with non-seminomatous germ cell tumor: a case report

Authors:

Muhammad N Zahir (nauman.zahir@aku.edu)
Nehal Masood (nehal.masood@aku.edu)
Munira S Moosajee (munira.moosajee@aku.edu)

Version: 2 Date: 22 June 2012

Author's response to reviews: see over
Dear Sir/Madam,

In this letter we are listing all the changes that have been made to our manuscript. These changes were made in order to accommodate the recommendations made by the editorial board. We hope that after these corrections, the manuscript conforms to the journal style of the *Journal of Medical Case Reports*.

1. The study design i.e. case report has been added to the title of the manuscript.
2. A title page has been added as page 1.
3. An abstract has been included on page 2 of the manuscript as per the required format.
4. The term gentleman has been replaced the term male to describe the patient’s gender.
5. The ethnicity of the patient has been included.
6. A conclusion section has been included as the last section of the text.
7. Consent statement has been revised.
8. A competing interests section has been added.
9. A section on the authors’ contributions has also been added.

We have made our best attempt at trying to address all the concerns of the editorial board. In case there are any further questions, please do not hesitate to contact us.

Thanking you,

Yours Sincerely,

Muhammad Nauman Zahir